University of Waterloo STAT 332 – W. H. Cherry

#2.105a

Figure 2.14. UNSTRATIFIED POPULATIONS: Estimating an Average or a Total Section headings to provide a content overview

One-Stage EPSWOR of Equal-Sized Clusters

1. Estimating Y−−, the Respondent Population Average

Example 2.14.1 Example 2.14.2 2. Efficiency

Example 2.14.3 Example 2.14.4 3. Appendix 1: Derivation of Equation (2.14.9) 4. Appendix 2: Standard Deviations 5. Appendix 3: Notation – Managing Falsehood 6. Acknowledgement Seven tables Two diagrams Thirteen equations

2023-08-20